Author's response to reviews

Title: Early stage signet ring cell carcinoma of the colon examined by magnifying endoscopy with narrow-band imaging: a case report

Authors:

Ken Ohnita (k-ohnita@nagasaki-u.ac.jp)
Hajime Isomoto (hajimei2002@yahoo.co.jp)
Taro Akashi (fuk_naga_ta23@yahoo.co.jp)
Keiichi Hashiguchi (cbpqc121@yahoo.co.jp)
Kayoko Matsushima (matsu7281978@yahoo.co.jp)
Hitomi Minami (le7novembre@gmail.com)
Yuko Akazawa (akazaway@nagasaki-u.ac.jp)
Naoyuki Yamaguchi (naoyuki3334@hotmail.com)
Fuminao Takeshima (ftake@nagasaki-u.ac.jp)
Kazuo To (tokazu@nagasaki-u.ac.jp)
Hiroaki Takeshita (takehiro@nagasaki-u.ac.jp)
Haruna Yasui (y-haruna@nagasaki-u.ac.jp)
Kuniko Abe (abek@nagasaki-u.ac.jp)
Kazuhiko Nakao (kazuhiko@nagasaki-u.ac.jp)

Version: 5
Date: 14 July 2015

Author's response to reviews: see over
14/07/2015  
Dr Elaine Zhang  
Editor  
*BMC Gastroenterology*

Dear editor  

Thank you for reviewing our manuscript and providing your comments and suggestions. We composed the figure images and uploaded 6 image files.

Sincerely,

Dr. Ken Ohnita  
Department of Gastroenterology and Hepatology  
Nagasaki University Hospital, 1-7-1 Sakamoto, Nagasaki, Japan.  
Phone No: +81-95-819-7481  
Fax No: +81-95-849-7482  
Email address: k-ohnita@nagasaki-u.ac.jp